%****************************************
%					*
%    MANUSCRIPIT for Proc ISBFS2006	*
%					*
%****************************************


% This file consists of 2 parts.

% (1) Please change the 2nd part.
% (2) Please do NOT change the 1st part.
% (3) If you define macros, please define them
%     between '\IsbfsDEFINEMACROS{' and '}'.
% (4) You can use '\IsbfsCITE{...}' instead of '\cite{...}'.


%****************************************
%					*
%	Part 1				*
%	Do not change the following	*
%					*
%****************************************


\newif\ifIsbfsINCLUDED
\ifx\IsbfsINCLUDED\undefined\IsbfsINCLUDEDtrue\else\IsbfsINCLUDEDfalse\fi
\ifIsbfsINCLUDED


\documentclass[11pt,leqno,twoside]{article}


\setlength{\paperheight}{224mm}
\setlength{\paperwidth}{156mm}
\setlength{\textwidth}{120mm}
\setlength{\textheight}{175mm}


\def\IsbfsUSEPACKAGE#1{\usepackage{#1}}


\makeatletter
\def\Isbfs@concat#1#2{%
 \expandafter\def\expandafter#1\expandafter{#1#2}}
\def\Isbfs@concat@b#1#2{%
 \expandafter\Isbfs@concat\expandafter#1\expandafter{#2}}


\pagestyle{myheadings}


\def\Isbfs@runningtitle{}
\def\Isbfs@title{}
\def\IsbfsTITLE[#1]#2{\def\Isbfs@runningtitle{#1}\def\Isbfs@title{#2}}


\def\IsbfsAUTHOR[#1]#2{%
\ifx\Isbfs@author@notlast\@empty
\Isbfs@concat@b\Isbfs@author@notlast\Isbfs@author@last
\Isbfs@concat@b\Isbfs@runauthor@notlast\Isbfs@runauthor@last\else
\Isbfs@concat\Isbfs@author@notlast{, }%
\Isbfs@concat@b\Isbfs@author@notlast\Isbfs@author@last
\Isbfs@concat\Isbfs@runauthor@notlast{, }%
\Isbfs@concat@b\Isbfs@runauthor@notlast\Isbfs@runauthor@last\fi
\def\Isbfs@author@last{#2}%
\def\Isbfs@runauthor@last{#1}%
\IsbfsAUTHOR@b#2\\ \@nil}

\def\IsbfsAUTHOR@b#1\\#2\@nil{%
\Isbfs@concat\Isbfs@alladdresses{\medskip\\{\sc #1}\\}}

\def\IsbfsADDRESS#1{\Isbfs@concat\Isbfs@alladdresses{#1\\}}

\def\Isbfs@author@notlast{}
\def\Isbfs@author@last{}
\def\Isbfs@runauthor@notlast{}
\def\Isbfs@runauthor@last{}
\def\Isbfs@alladdresses{}

\def\Isbfs@MAKEauthors{%
 \def\Isbfs@authors{}\def\Isbfs@runauthors{}%
 \ifx\Isbfs@author@notlast\@empty
 \Isbfs@concat@b\Isbfs@author@notlast\Isbfs@author@last
 \Isbfs@concat@b\Isbfs@runauthor@notlast\Isbfs@runauthor@last\else
 \Isbfs@concat\Isbfs@author@notlast{ and }%
 \Isbfs@concat@b\Isbfs@author@notlast\Isbfs@author@last
 \Isbfs@concat\Isbfs@runauthor@notlast{ and }%
 \Isbfs@concat@b\Isbfs@runauthor@notlast\Isbfs@runauthor@last\fi
 \expandafter\Isbfs@MAKEauthors@b\Isbfs@author@notlast\\,\@nil
 \Isbfs@concat@b\Isbfs@runauthors\Isbfs@runauthor@notlast}

\def\Isbfs@MAKEauthors@b#1\\,#2\@nil{%
 \def\Isbfs@authors@b{#2}%
 \ifx\Isbfs@authors@b\@empty
 \Isbfs@concat\Isbfs@authors{#1}\else
 \Isbfs@concat\Isbfs@authors{#1,\\}%
 \expandafter\Isbfs@MAKEauthors@b#2\@nil\fi}

\def\IsbfsEMAIL#1{%
\Isbfs@concat\Isbfs@alladdresses{Email: {\bf #1} \\}}


\def\IsbfsKEYWORDS#1{\def\Isbfs@keywords{#1}}
\def\Isbfs@keywords{}
\def\IsbfsSUBJCLASS#1{\def\Isbfs@subjclass{#1}}
\def\Isbfs@subjclass{}
\def\IsbfsTHANKS#1{\def\Isbfs@thanks{#1}}
\def\Isbfs@thanks{}
\def\IsbfsDEDICATEDTO#1{\def\Isbfs@dedicatedto{#1}}
\def\Isbfs@dedicatedto{}

\long\def\IsbfsABSTRACT#1{\long\def\Isbfs@abstract{#1}}


\def\IsbfsMAKETITLE{
\thispagestyle{plain}
\Isbfs@MAKEauthors
\markboth{\Isbfs@runauthors}{\Isbfs@runningtitle}

\begin{center}
\large\bf
\Isbfs@title
\end{center}

\begin{center}
\Isbfs@authors
\end{center}
\Isbfs@definemacros

\ifx\Isbfs@dedicatedto\@empty\smallskip\else
\begin{center}\Isbfs@dedicatedto\end{center}\fi

\begin{quote}
{\sc Abstract.}
\Isbfs@abstract
\ifx\Isbfs@subjclass\@empty\relax\else
{\def\thefootnote{{}}%
\footnote{2000 {\it subject classification}: \Isbfs@subjclass}%
\addtocounter{footnote}{-1}}\fi
\ifx\Isbfs@keywords\@empty\relax\else
{\def\thefootnote{{}}%
\footnote{{\it Key words and phrases}: \Isbfs@keywords}%
\addtocounter{footnote}{-1}}\fi
\ifx\Isbfs@thanks\@empty\relax\else
{\def\thefootnote{{}}%
\footnote{\Isbfs@thanks}%
\addtocounter{footnote}{-1}}\fi
\end{quote}
}


\long\def\IsbfsDEFINEMACROS#1{\long\def\Isbfs@definemacros{#1}}


\def\bysame{\leavevmode\hbox to3em{\hrulefill}\thinspace}

\def\IsbfsBIBLIOGRAPHY{
\begin{thebibliography}{99}
\itemsep=6pt plus 1pt}

\def\endIsbfsBIBLIOGRAPHY{
\end{thebibliography}}

\def\IsbfsDOCUMENT{\document\IsbfsMAKETITLE}
\def\Isbfs@newline{\par\noindent}
\def\endIsbfsDOCUMENT{%
\begingroup\let\\=\Isbfs@newline\Isbfs@alladdresses\endgroup
\enddocument}


% Suzuki-cite ver 1.03
\newcounter{Suzuki@citeA}
\newcounter{Suzuki@citeB}
\newcounter{Suzuki@citeC}
\newcounter{Suzuki@citeD}
\def\IsbfsCITE#1{%
 \begingroup[\setcounter{Suzuki@citeA}{-1}\Suzuki@Cite{#1}]\endgroup}
\def\Suzuki@Cite#1{%
 \let\Suzuki@Cite@a\@empty
 \let\Suzuki@Cite@b\@empty
 \@for\Suzuki@Cite@i:=#1\do{%
 \edef\Suzuki@Cite@i{\expandafter\@firstofone\Suzuki@Cite@i\@empty}%
 \@ifundefined{b@\Suzuki@Cite@i}{%
  \Suzuki@Cite@b\let\Suzuki@Cite@b\@empty
  \Suzuki@Cite@a
  \setcounter{Suzuki@citeA}{-1}%
  \mbox{\reset@font\bfseries ?}%
  \G@refundefinedtrue
  \@latex@warning{Citation `\Suzuki@Cite@i' on page \thepage \space
    undefined}}%
 {\setcounter{Suzuki@citeB}{\csname b@\Suzuki@Cite@i\endcsname}%
  \ifnum\value{Suzuki@citeA}=\value{Suzuki@citeB}%
   \setcounter{Suzuki@citeD}{\the\value{Suzuki@citeB}}%
   \ifnum\value{Suzuki@citeC}<\value{Suzuki@citeB}%
    \def\Suzuki@Cite@b{--\Suzuki@Cite@Form{\the\value{Suzuki@citeD}}}%
   \else\def\Suzuki@Cite@b{,\penalty\@m
    \ \Suzuki@Cite@Form{\the\value{Suzuki@citeD}}}%
   \fi\addtocounter{Suzuki@citeA}{1}%
  \else
   \Suzuki@Cite@b\let\Suzuki@Cite@b\@empty
   \Suzuki@Cite@a
   \hbox{\Suzuki@Cite@Form{\the\value{Suzuki@citeB}}}%
   \setcounter{Suzuki@citeA}{\the\value{Suzuki@citeB}}%
   \addtocounter{Suzuki@citeA}{1}%
   \setcounter{Suzuki@citeC}{\the\value{Suzuki@citeA}}%
  \fi}%
 \def\Suzuki@Cite@a{,\penalty\@m\ }}%
 \Suzuki@Cite@b}
\def\Suzuki@Cite@Form#1{#1}


\def\IsbfsINDENTAFTERSECTION{
\IsbfsINDENTAFTERSECTION@b\section\Isbfs@SAVE@SECTION
\IsbfsINDENTAFTERSECTION@b\subsection\Isbfs@SAVE@SUBSECTION
\IsbfsINDENTAFTERSECTION@b\subsubsection\Isbfs@SAVE@SUBSUBSECTION
}

\def\IsbfsINDENTAFTERSECTION@b#1#2{
\let#2#1
\def#1{\@ifstar{\IsbfsINDENTAFTERSECTION@c#2}%
{\IsbfsINDENTAFTERSECTION@d#2}}}

\def\IsbfsINDENTAFTERSECTION@c#1#2{#1*{#2}\@afterindenttrue}

\def\IsbfsINDENTAFTERSECTION@d#1#2{#1{#2}\@afterindenttrue}


\makeatother
\fi


%****************************************
%					*
%	Part 2				*
%	Please change the following	*
%					*
%****************************************


% \IsbfsUSEPACKAGE{amssymb,amsmath} % please delete the fisrt '%' if you use
% \IsbfsUSEPACKAGE{amsfonts,amsthm}


% You cannot use your macros in `Title' and `Author(s)'.
\IsbfsTITLE[running title]{Type, cotype and convexity
properties\\ of quasi-Banach spaces}

\IsbfsAUTHOR[S. Banach]{Stefan Banach}
\IsbfsADDRESS{Department of Mathematics,
Kyushu Institute of Technology,
Tobata, Kitakyushu 804-8550, Japan}
\IsbfsEMAIL{banach@mns.kyutech.ac.jp}

  \IsbfsAUTHOR[G. Hijkl]{2nd author}
% \IsbfsAUTHOR[G. Hijkl]{2nd author\\}	% if a `newline' is needed
\IsbfsADDRESS{postal address of 2nd author}
\IsbfsEMAIL{e-mail address of 2nd author}

\IsbfsAUTHOR[M. Nopqr]{3rd author}
\IsbfsADDRESS{postal address of 3rd author}
\IsbfsEMAIL{e-mail address of 3rd author}

\IsbfsTHANKS{The first and the third authors are supported by ....}
% \IsbfsDEDICATEDTO{Dedicated to ....} % if you use




\IsbfsDEFINEMACROS{

% If you define macros, please define them here.
% \newcommand...{...}

% You can change the following macros to your own ones.
\newtheorem{thm}{Theorem}
\newtheorem{lem}[thm]{Lemma}
\newtheorem{cor}[thm]{Corollary}
\newtheorem{ack}{Acknowledgment\ignorespaces}
\renewcommand{\theack}{}

% \IsbfsINDENTAFTERSECTION % if you need an indent after '\section'.
% \let\cite\IsbfsCITE % if you use '\cite' as '\IsbfsCITE'.

% \newcommand{\abcdefg}[2]{abcdefg ##1 abcdefg ##2 abcdefg}
} % end of \IsbfsDEFINEMACROS




\IsbfsSUBJCLASS{Primary 12A34, Secondary 56B78, 90C12}
\IsbfsKEYWORDS{Banach space, Function space, Uniform convexity}


\IsbfsABSTRACT{%
Our abstract ....
}


\begin{IsbfsDOCUMENT}


\section{Introduction}

aaa

aaa


[3--5] ... IsbfsCITE
% \IsbfsCITE{A, B, C} ... IsbfsCITE

\cite{A, B, C} ... originnal cite

\begin{thm}[Megginson \cite{Megginson}]
\label{X}
Theorem
\end{thm}

In Theorem \ref{X}, ...


\begin{lem}
\label{Z}
Lemma
\end{lem}

By Lemma \ref{Z}, ...


\begin{cor}
\label{Y}
Corollary
\end{cor}


a

b

c

d

e

f

g

h

i

j

k

l

m

n

o

p

q

r

s

t

u

v

w

x

y

z

a

b

c

d

e

f

g

h

i

j

k

l

m

n

o

p

q

r

s

t

u

v

w

x

y

z


\begin{ack}\rm
Acknowledgment
\end{ack}


\begin{IsbfsBIBLIOGRAPHY}

\bibitem{Gowers-Maurey}
W. T. Gowers and B. Maurey,
{\it The unconditional basic sequence problem},
J.\ Amer.\ Math.\ Soc.\ {\bf 6} (1993), 851--874.

\bibitem{Megginson}
R. E. Megginson,
{\it An introduction to Banach space theory},
Springer-Verlag, New York, 1998.

\bibitem{A} author, title, journal...

\bibitem{B} author, title, journal...

\bibitem{C} \bysame, title, journal...

\end{IsbfsBIBLIOGRAPHY}


\end{IsbfsDOCUMENT}
